Author’s response to reviews

Title: Epidemiology, associated factors and outcomes of ICU-acquired infections caused by Gram-negative bacteria in critically ill patients: an observational, retrospective study

Authors:

Cosimo Chelazzi (cosimochelazzi@gmail.com)
Eleonora Pettini (ele.pet@gmail.com)
Gianluca Villa (gianlucavilla1@gmail.com)
A. Raffaele De Gaudio (araffaele.degaudio@unifi.it)

Version: 5

Date: 19 July 2015

Author’s response to reviews:

Dear Editor of BMC Anesthesiology,

thank you for your interest in our paper.

We changed the manuscript according to the editorial revisions you required us to make.

In the “abstract” section we changed “introduction” to “background”

Line 27-28: we provided the full name and affiliation of the Ethics Committee that approved the study

Line 287 (now 261, “competing interests”): we changed “author(s)” to “authors”

We formatted the revised manuscript according to the journal style.

Thank you very much,

Cosimo Chelazzi,

on behalf of all authors